Author's response to reviews

Title: Immunohistochemical detection improves the prognostic value of lymphatic and blood vessel invasion in primary ductal breast cancer

Authors:

Fadia JA Gujam (f.gujam.1@research.gla.ac.uk)
James J Going (Games.Going@glasgow.ac.uk)
Zahra MA Mohammed (mrxzma@hotmail.com)
Clare Orange (clare.orange@glasgow.ac.uk)
Joanne Edwards (Joanne.Edwards@glasgow.ac.uk)
Donald C McMillan (Donald.McMillan@glasgow.ac.uk)

Version: 8 Date: 9 September 2014

Author's response to reviews: see over
09/09/ 2014  
BioMed Central Editorial,

Dear Dr Britta Weigelt,

MS: 2099120226114527 - Immunohistochemical detection improves the prognostic value of lymphatic and blood vessel invasion in primary ductal breast cancer.

Thank you for your recent e-mail. We acknowledge the comments of the editor and we have considered them carefully and these have improved the manuscript. There follows our response;

1. "We strongly encourage you to include an Acknowledgements section between the Authors' contributions section and Reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study.

Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements. Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication."

Point taken. We have now added an acknowledgements section between the Authors' contributions section and Reference list as follow;

"Acknowledgements

The authors gratefully acknowledge funding from Libyan government- Ministry of Higher Education."

Yours sincerely
Fadia JA Gujam
Donald C McMillan